Author's response to reviews

Title: 'Reaching the Hard to Reach' - Lessons learned from the VCS (Voluntary and Community Sector)

Authors:

Sarah M Flanagan (s.m.flanagan@bham.ac.uk)
Beverley Hancock (b.hancock@bham.ac.uk)

Version: 2 Date: 17 December 2008

Author's response to reviews:

Dear Mark,

Apologies for inadequate submission details.

I am resubmitting the paper with the following revisions;

1. Title Page
2. Abstract
3. Competing Interest and Authors' contributions section
4. Structure and content revision.

Many thanks

Sarah Flanagan